Author’s response to reviews

Title: Transcutaneous Neuromodulation improved inflammation and sympathovagal ratio in Patients with Primary Biliary Cholangitis and Inadequate Response to Ursodeoxycholic Acid: A Pilot Study

Authors:

Hui Yang (602474759@qq.com)
Hang Yang (15841123892@163.com)
Lixia Wang (wenwensl@yeah.net)
Honggang Shi (1666520879@qq.com)
Bojia Liu (715493908@qq.com)
Xue Lin (1987577265@qq.com)
Qingyong Chang (qychang0409@163.com)
Jiande Chen (jiandedzchen@gmail.com)
Zhijun Duan (cathydoctor@sina.com)

Version: 5  Date: 23 Jul 2020

Author’s response to reviews:

Dear editor,
It is a great honor for us to have the chance of publication in the journal. And many thanks again to the very kind and responsible editors and reviewers.

1. Please include the full name of the ethics committee (and the institute to which it belongs) that approved the study under the Ethics approval and consent to participate heading. You have not specified which hospital in your current statement.

The study protocol was approved by the ethics committee of the First Affiliated Hospital of Dalian Medical University (number PJ-KY-2017-113(X))

2. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The fund had no role in the design of the study, collection, analysis, interpretation of data, or writing the manuscript.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

The clean version of the manuscript and figure legends has been uploaded.